Author’s response to reviews

Title: Molecular Identification and Antifungal Susceptibility Profile of Aspergillus flavus Isolates Recovered from Clinical Specimens in Kuwait

Authors:

Faten A Al-Wathiqi (fa10_1987@hotmail.com)
Suhail Ahmad (suhail_ah@hsc.edu.kw)
Ziauddin Khan (zkhan@hsc.edu.kw)

Version: 4 Date: 20 February 2013

Author’s response to reviews:

Title: Molecular identification and Antifungal Susceptibility Profile of Aspergillus flavus Isolates Recovered from Clinical Specimens in Kuwait

Dr. Philippa Harris
Executive Editor
BMC-series Journals
BioMed Central

Dear Dr. Harris,

Many thanks for forwarding us the comments of the reviewer.

As suggested, we have incorporated additional information (highlighted in red font) on antifungal susceptibility pattern of A. flavus isolates recovered over an 18-year period.

We, sincerely hope that you will find the revised manuscript suitable for publication in BMC Infectious Diseases.

Looking forward to hearing from you.

Best regards,

Z. U. Khan